**Author's response to reviews**

**Title:** Factors associated with early menarche: results from the French Health Behaviour in School-aged Children (HBSC) study

**Authors:**

Adrien Gaudineau (a.gaudineau@orange.fr)
Virginie Ehlinger (virginie.ehlinger@yahoo.fr)
Christophe Vayssiere (c.vayssiere@club-internet.fr)
Beatrice Jouret (jouret.beatrice@chu-toulouse.fr)
Catherine Arnaud (carnaud@cict.fr)
Emmanuelle Godeau (emmanuelle.godeau@ac-toulouse.fr)

**Version:** 3  **Date:** 24 March 2010

**Author's response to reviews:**

The authors have changed the abstract in the submission system in order to match the abstract in the manuscript.
The authors have changed the title in sentence case.
The authors have removed the authors' qualifications from the manuscript.
The authors have removed the paragraph in the manuscript "the authors think..."